Author's response to reviews

Title: Relationships between low serum Vitamin D3, BMI and Waist in Overweight and Obesity

Authors:

Anne-Thea McGill (at.mcgill@auckland.ac.nz)
Joanna M Stewart (j.stewart@auckland.ac.nz)
Fiona E Leahy (f.leahy@auckland.ac.nz)
Caroline Strik (c.strik@auckland.ac.nz)
Sally D Poppitt (s.poppitt@auckland.ac.nz)

Version: 2 Date: 21 June 2006

Author's response to reviews:

The formatting of the Table 1 caption, using the Word "footnote" function, has become an error note. Footnotes have been removed and numbers added manually.